Author's response to reviews


Authors:

Eva Dannetun (eva.dannetun@lio.se)
Anders Tegnell (anders.tegnell@socialstyrelsen.se)
Johan Giesecke (johan.giesecke@ecdc.eu.int)

Version: 3 Date: 24 April 2007

Author's response to reviews: see over
Dear Editor

We would like to thank your referees for constructive criticism of our manuscript, Parents attitudes towards hepatitis B vaccination for their children. A survey comparing paper and web questionnaires, Sweden 2005. We have tried to incorporate the comments as follows:

Referee 1,
Major Compulsory Revisions
The reviewer correctly points out that in the main analysis it was unclear which were the independent and which were the dependent variables. We have added a sentence in 'Results' clarifying this, and also changed the heading of Table 3 slightly, (p.8, under the subtitle: Knowledge and attitudes, line 2-9).

Minor points
All remarks under the heading Minor Essential Revisions, have been agreed upon and altered.

We see that our article has improved from these alterations.

Linköping the 25th of April 2007
Yours sincerely,
Eva Dannetun